Author's response to reviews

Title: Measuring capacity building in communities: a review of the literature

Authors:

Selma C Liberato (selma.liberato@menzies.edu.au)
Julie Brimblecombe (Julie.brimblecombe@menzies.edu.au)
Jan Ritchie (j.ritchie@unsw.edu.au)
Megan Ferguson (megan.ferguson@menzies.edu.au)
John Coveney (john.coveney@flinders.edu.au)

Version: 4 Date: 4 November 2011

Author's response to reviews: see over
Minor corrections in:

- affiliations;

- Some grammatical corrections in table 3;

- some grammatical corrections in the main document;

- inclusion of some "quotes" for the domains;

- some references (3, 12, 48, 59 and 61) were corrected e.g. some of them had the authors initial names missing.